Author's response to reviews

Title: Effect of Tualang honey on the healing of anastomotic wound in large bowel anastomosis in rats-A randomized controlled trial

Authors:

Muhammad Izani Aznan (alnerazzuri@yahoo.com)
Omaid Hayat Khan (omaidhayatkhan@hotmail.com)
Allah Obhayo Unar (unarobhayo@yahoo.com)
Sharifah Emilia Tuan Sharif (dremilia@usm.my)
Amer Hayat Khan (amerhayat@ymail.com)
Syed Hassan (shassan@usm.my)
Andee Dzulkarnain Zakaria (andee778@gmail.com)

Version: 2
Date: 27 April 2015

Author's response to reviews:

Dear Editor,

Few revisions were asked which have been revised and uploaded in the newer version. Details of which are as follows:

1) Line and page numbering were asked which have been added accordingly.
2) Removed all the Page Breaks so as to ensure the consistent page numbering.
3) Source of Tualang Honey has been updated in the Methods Section.
4) Acknowledgment section has been updated as well with the details of the research grant.

Thanks and regards,